Author’s response to reviews

Title: Testicular Tuberculosis presenting with metastatic intracranial tuberculomas only: A case report.

Authors:

Godwin I Ogbole (gogbole@yahoo.com)
Oku S Bassey (tettebaby_tl@yahoo.com)
Clement A Okolo (clemabuokolo@yahoo.com)
Samson O Ukperi (uksamsonik@yahoo.com)
Ayotunde O Ogunseyinde (remioguns2001@yahoo.com)

Version: 7 Date: 19 November 2010

Author’s response to reviews:

Dear Editor,

The following changes have been effected as requested.

1. Email addresses: the email address of Oku S Bassey on the manuscript has been corrected and now identical with that on the submission system. Also the format using author initials has been adopted.

2. References: we have ensured that the references are now in numerical order.

3. Coloured text: All text is now presented in black.

4. Structure: the competing interests section is now before the consent section and the figure legends has been moved to the end of the manuscript, after the references.

5. Typography: we have used this opportunity to check the manuscript for typographical errors.

Thank you.